Author's response to reviews

Title: Primary irritant and delayed-contact hypersensitivity reactions to the freshwater cyanobacterium Cylindrospermopsis raciborskii and its associated toxin cylindrospermopsin

Authors:

Ian Stewart (i.stewart@uq.edu.au)
Alan A Seawright (a.seawright@uq.edu.au)
Philip J Schluter (philip.schluter@aut.ac.nz)
Glen R Shaw (g.shaw@griffith.edu.au)

Version: 4 Date: 27 March 2006

Author's response to reviews:

Bibliography updated for publications by Stewart et al.

Can BMC production please update citation details for reference #9 when published (that paper is a sister paper to this one, to be published in the same volume)?